

November 25, 2014

Dear Editor,

Please find enclosed the edited manuscript for the World Journal of Gastroenterology

Title: Cobalamin deficiency as an extra intestinal manifestation of *Helicobacter pylori*

Author: Emmanuel Andrès

Name of Journal: *World Journal of Gastroenterology*

ESPS Manuscript NO: ESPS Manuscript NO: 14043

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 Revision has been made according to the suggestions of the reviewer

(1) A running title was added.

(2) References were accorded to the standard format of the Journal.

3 References and typesetting were corrected.

The paper was reviewed and edited by the Dr Helen Fotherghill (English native speaker, pediatrics department, saint thomas hospital in Lindon).

Thank you again for publishing our manuscript in the *World Journal of Gastroenterology*.

Sincerely yours,

Pr Emmanuel ANDRES

Professeur des Universit és – Praticien Hospitalier

Professeur de M édecine Interne

Facult é de M édecine - Universit é de Strasbourg

Chef de Service

Service de M édecine Interne, Diab ète et Maladies m étaboliques

Clinique m édicale B - CHRU de STRASBOURG

Tel : 33388115066

Fax : 33388116262

Mail : emmanuel.andres@chru-strasbourg.fr